Author's response to reviews

Title: Renal dysfunction, restrictive left ventricular filling pattern and mortality risk in patients admitted with heart failure: A 7-year follow-up study

Authors:

Morten Schou (m.schou@dadlnet.dk)
Jesper Kjargaard (m.schou@dadlnet.dk)
Chr Torp-pedersen (m.schou@dadlnet.dk)
Finn Gustafsson (m.schou@dadlnet.dk)
Lars Kober (m.schou@dadlnet.dk)
Chr Hasager (m.schou@dadlnet.dk)
Jacob Moller (m.schou@dadlnet.dk)
Dilek Akkan (m.schou@dadlnet.dk)

Version: 2 Date: 29 October 2012

Author's response to reviews:

Revisions according to email

Sincerely yours,

Morten Schou